
\magnification1200\footline{}
\centerline{\bf Math.~126 Quizz \#9}
\centerline{April 6, 1999}
\vskip10pt
\noindent

\def\frac#1#2{{{\textstyle#1}\over{\textstyle#2}}}
\def\isum#1{{\displaystyle{\sum_{n={#1}}^{\infty}}}\>}

\noindent
Give one example of each of the following:
\itemitem{(a)} a divergent series
\itemitem{(b)} a conditionally convergent series
\itemitem{(c)} an absolutely convergent series.

\noindent
For each of your examples, give a brief reason why it is an
example.

\end

Sent Wed Mar 31 15:44:15 EST 1999
